Author's response to reviews

Title: Effects of Mothers' Socio-Economic Status on the Management of febrile conditions in their under five children in a Resource Limited Setting.

Authors:

Adenike E Olaogun (nikeolaogun@yahoo.com)
Abayomi A Adebayo (bayoade@oauife.edu.ng)
Olufemi E Ayandiran (olufeman@yahoo.com)
Olayinka A Olasode (olayinkaolasode@yahoo.com)

Version: 11 Date: 15 December 2005

Author's response to reviews:

POINT BY POINT COMMENT.

Reviewer: Girdhar AGARWAL

Comment 1;
The request for the use of 'Logistic Regression' instead of 'Stepwise Regression'.

Response;
This has being effected in the manuscript. Binary logistic regression was used instead of Step wise regression.

Comment 2;
The need for table 3 to be reframed.

Response;
Table 3 has been revised to reflect the results of Binary Logistic Regression.

Comment 3;
Use of mean and standard deviation.

Response;
The mean and standard deviation were used in the manuscript.